Author's response to reviews

Title: Modern contraceptive utilization among female ART attendees in health facilities of Gimbie town, West Ethiopia

Authors:

Addisu Polisi (addispolis@gmail.com)
Ewenat Gebrehanna (gewenat@gmail.com)
Gezahegn Tesfaye (gezites@gmail.com)
Fekede Asefa (sinboona@gmail.com)

Version: 3
Date: 28 March 2014

Author's response to reviews: see over
Author's reply to Editors comments

Title: Modern contraceptive utilization among female ART attendees in health facilities of Gimbie town, West Ethiopia

Authors:

Addisu Polisi: addispolis@gmail.com
Ewenat Gebrehanna: gewenat@gmail.com
Gezahegn Tesfaye: gezites@gmail.com
Fekede Asefa: sinboona@gmail.com

Version: 3, Date: March 28, 2014

To: Jose Belizan, Editor-in-Chief, the Reproductive Health Editorial Team
From: Gezahegn Tesfaye (Co-author)

Title: Modern contraceptive utilization among female ART attendees in health facilities of Gimbie town, West Ethiopia

Subject: Comments incorporated

Thank you so much for the comments. Below in the table you will find how the comments are considered in the modification of the manuscript.

A. Revisions

<table>
<thead>
<tr>
<th>Editor's comments</th>
<th>Authors reply</th>
</tr>
</thead>
<tbody>
<tr>
<td>Conclusion/possible action part</td>
<td>Thank you so much. We have deleted unnecessary recommendation and come up with possible action that was based on the findings of the study.</td>
</tr>
</tbody>
</table>

I am asking you to delete the section Possible Actions. Instead of this if you wish you can summarize these recommendations in a sentence only mentioning the recommendations that are derived from results obtained in this study.
<table>
<thead>
<tr>
<th><strong>Reference part</strong></th>
</tr>
</thead>
<tbody>
<tr>
<td>Please, you need to strongly work on the references. The majority of them cannot be located. There are many of them without describing the Journal, many of them without a link to the web page where they are obtained. Please see also that our Journal requires the list of authors instead of mentioning only the first author. Please follow the instructions about references in the author instructions of the Journal.</td>
</tr>
</tbody>
</table>

| |
| Thank you so much. We have addressed those concerns in the reference part. We have work out on the reference part according to the journal referencing style. |